Author's response to reviews

Title: Incorporating natural variation into IVF clinic league tables: The Expected Rank

Authors:

Hester F Lingsma (h.lingsma@erasusmc.nl)
Marinus JC Eijkemans (m.eijkemans@erasusmc.nl)
Ewout W Steyerberg (e.steyerberg@erasusmc.nl)

Version: 6 Date: 14 July 2009

Author's response to reviews:

Ref 10 is removed and now appears in the text, since it is a paper under revision, and not yet published